Author's response to reviews

Title: Intramedullary neurocysticercosis presenting as Brown Sequare Syndrome

Authors:

Elda M Salazar (massiels24@gmail.com)
Fernando J Escoto (escotomd@gmail.com)
Rita Pineda (reapineda64@gmail.com)

Version: 2 Date: 26 August 2014

Author's response to reviews:

Dear Editor:

The changes made to the previous manuscript have been the following:

- A title page at the front of the manuscript was included.
- Page numbers were included.

Thank you,

Elda Salazar, MD.
Hospital General San Juan de Dios
Guatemala, Guatemala
massiels24@gmail.com
phone: 50257526564